Author’s response to reviews

Title: Management Of Arrhythmogenic Right Ventricular Dysplasia In Pregnancy: A Rare Case

Authors:

Nilgün Güdüçü (nilgun.kutay@gmail.com)
Salih S Kutay (serdarkutay@hotmail.com)
Ebru Özenç (eozenc@hotmail.com)
Çavlan Çiftçi (cavlanciftci@superonline.com)
Alin B Yi (&#287;iter (alinbasgul@gmail.com)
Herman &#304;&#351;i (hermani@superonline.com)

Version: 4 Date: 16 February 2011

Author’s response to reviews:

Dear Editor,
We have provided the suggested changes:
- An abstract is added to manuscript on page 2
- Consent is revised
- A header is added to conclusion
- Ethnicity of the patient is written within the case presentation
Thank you very much for your collaborations,
Best regards,
Nilgün Güdüçü